Author's response to reviews

Title: Factors associated with utilization of long acting and permanent contraceptive methods among married women of reproductive age in Mekelle town, Tigray region north Ethiopia

Authors:

Mussie Alemayehu (mossalex75@gmail.com)
Tefera Belachew (tefera_belachew@yahoo.com)
Tizta Tilahun (tiztatilahun@yahoo.com)

Version: 9 Date: 25 January 2012

Author's response to reviews:

References were ordered and table number four re-arranged in a new way.